Author's response to reviews

Title: Pyogenic brain abscess, a 15 year survey

Authors:

Jannik Helweg-Larsen (jhelweg@dadlnet.dk)
Arnar Astradsson (arnar.astradsson@gmail.com)
Humeira Richhall (richhall786@yahoo.dk)
Jesper Erdal (erdal@dadlnet.dk)
Alex Laursen (d201790@dadlnet.dk)
Jannick Brennum (jannick.brennum@rh.regionh.dk)

Version: 2 Date: 25 June 2012

Author's response to reviews:

Dear Sirs

As requested- the following small changes and amendments have been made:

1. Please confirm that permission was obtained to access patient records for your study.

   The following sentence has now been included in the Method section, page 4:
   "Permission to access patient records was obtained in accordance with the Danish Act on processing of personal data and the study was approved by the Copenhagen ethics committee."

2. Request for competing interest section:

   This has now been included on page 12:
   "Competing interests: The authors declare that they have no financial or non-financial competing interests."

3. Request for acknowledgments section.

   Section now included- on page 12:
   "Acknowledgements: Professor Peter Skinhøj for advice and support. "